Author's response to reviews

Title: Outcomes of patients hospitalized for acute decompensated heart failure: does nesiritide make a difference?

Authors:

Richard J Carroll (richard.carroll@ahss.org)
Zuber D Mulla (zuber.mulla@ttuhsc.edu)
Loran D Hauck (loran.hauck@ahss.org)
Audrey Westbrook (audrey.westbrook@fhmd.org)

Version: 4 Date: 20 November 2007

Author's response to reviews:

Dear Staff:

This paper was accepted recently. I am uploading the final version with the requested formatting changes.

Thank you,
Prof. Zuber Mulla